Author's response to reviews

Title: Screening and selection of peptides specific for esophageal cancer cells from a p

Authors:

Zhe-Feng Zhang (doctorzzf@163.com)
Xue Shan (shanxue2046@126.com)
Yong-Xin Wang (251473841@qq.com)
Wei Wang (lianniba521@qq.com)
Shi-Yun Feng (871947169@qq.com)
You-Bin Cui (doctorcyb@163.com)

Version: 4
Date: 16 April 2014

Author's response to reviews: see over
Author's response to reviews

Title: Peptides specific for esophageal cancer cells were selected from a phage display peptide library

Authors:
Zhe-Feng Zhang, 
Xue Shan, 
Yong-Xin Wang, 
Wei Wang, 
Shi-Yun Feng, 
You-Bin Cui

Version: 2
Date: 14-April-2014
Author's response to reviews: see over

Reviewer's report

Title: Peptides specific for esophageal cancer cells were selected from a phage display peptide library

Version: 3 Date: 28 March 2014
Reviewer: Harry Xia Xia

Reviewer's report:
The authors have responded to my comments and revised the manuscript to my satisfaction. Please consider further modifying the title as "Screening and selection of peptides specific for esophageal cancer cells from a phage display peptide library".

The title has been modified according to the reviewer's suggestion.

Level of interest: An article of outstanding merit and interest in its field

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
No for all above.